Author's response to reviews

Title: Modification of stool's water content in constipated infants: management by an adapted infant formula

Authors:

Dámaso D Infante (dinfante@vhebron.net)
Oscar O Segarra (osegarra@vhebron.net)
Susana S Redecillas (sredecil@vhebron.net)
Marina M Alvarez (maralvarez@vhebron.net)
Mar M Miserachs (mar@miserachs.com)

Version: 2 Date: 28 July 2010

Author's response to reviews:

Suggested modifications have been performed without significant modifications in the main text.